Author’s response to reviews

Title: Performance management methods and practices among Nurses in Primary Health Care Settings: A systematic scoping review protocol

Authors:

Cynthia Madlabana (madlabana@ukzn.ac.za;zmadlabane@gmail.com)

Tivani Mashamba-Thompson (mashamba-thompson@ukzn.ac.za)

Inge Petersen (Peterseni@ukzn.ac.za)

Version: 2 Date: 01 May 2019

Author’s response to reviews:

Thank you, your feedback is valuable to us.

The manuscript has been reworked. The track changes have been retained so that the reviewers can see the changes.

1. The abstract, background and discussion section has been reworked to address reviewers comments. Please see reworked sections.

2. Some acronyms were added without providing the full name first. This was corrected throughout the manuscript.

3. According to the guideline, authors should minimize the use of abbreviations in the abstract. Only two abbreviations remain in the abstract, these are PM and PHC

All specific comments have been addressed adequately.